Author's response to reviews

Title: Virological pattern of HBV infection in an HIV Positive Patient with a Fatal HBV-related fulminant hepatitis (FHB): a case report.

Authors:

Sabrina Bagaglio (bagaglio.sabrina@hsr.it)
Luca Albarello (albarello.luca@hsr.it)
Priscilla Biswas (biswas.priscilla@hsr.it)
Caterina Uberti-Foppa (uberti.caterina@hsr.it)
Claudio Fortis (fortis.claudio@hsr.it)
Giulia Morsica (morsica.giulia@hsr.it)

Version: 8 Date: 21 January 2009

Author’s response to reviews: see over
Milan, 21-01-09

To

Journal of Medical Case Reports, Editorial Team

BioMed Central

Middlesex House

34-42 Cleveland Street, London W1T 4LB

+44 (0) 207 323 0 323

MS: 1718129002049705

We re-submit a revised version of the manuscript (text) entitled: “VIROLOGICAL PATTERN OF HBV INFECTION IN AN HIV-POSITIVE PATIENT WITH FATAL HBV-RELATED FULMINANT HEPATITIS (FHB): A CASE REPORT”, by Bagaglio et al. and cover letter in accordance with the comment of the Associate Editor.

In detail, a sentence regarding the recent sexual history of the patient (last two years) was added in the section ‘Case Presentation’ page 3, lines 13,14 from the top.

We made every effort in revising the manuscript and we hope it will be now suitable for publication in the JMCR.

Sincerely yours,

Giulia Morsica

Giulia Morsica
Infectious Diseases Dept.,
San Raffaele, Scientific Institute, Via Stamira d’Ancona 20,
20127 Milano, Italy.
Tel: +39-02-26437935; fax: +39-02-26437989; e-mail: morsica.giulia@hsr.it